Author's response to reviews

Title: Target enzyme mutations are the molecular basis for resistance towards pharmacological inhibition of nicotinamide phosphoribosyltransferase

Authors:

Uffe H Olesen (olesen.uh@gmail.com)
Jakob G Petersen (jakobgramstrup@gmail.com)
Antje Garten (antje.garten@medizin.uni-leipzig.de)
Wieland Kiess (wieland.kiess@medizin.uni-leipzig.de)
Jun Yoshino (yoshinojun@wustl.edu)
Shin-Ichiro Imai (imaishin@wustl.edu)
Mette K Christensen (mkc@topotarget.com)
Peter Fristrup (peter.fristrup@gmail.com)
Annemette V Thougaard (AVTH@Lundbeck.com)
Fredrik Björkling (fb@farma.ku.dk)
Peter B Jensen (pbj@topotarget.com)
Søren J Nielsen (sje@exiqon.com)
Maxwell Sehested (mxs@topotarget.com)

Version: 6 Date: 22 November 2010

Author's response to reviews:

Final revisions:

Abstract has been copied to be identical in the manuscript and in the submission system.

Figure 2 has been issued an overall title.

"10.9 ± 6.1" gone missing from table 2 during formatting has been reintroduced.